Author’s response to reviews

Title: Easier said than done: challenges of applying the Ecohealth approach to the study on heavy metals exposure among indigenous communities of the Peruvian Amazon

Authors:

Cynthia Anticona Huaynate (cynthiaanticona@yahoo.com)
Anna-Britt Coe (anna-britt.coe@ucgs.umu.se)
Ingvar A Bergdahl (ingvar.bergdahl@envmed.umu.se)
Miguel San Sebastian (miguel.sansebastian@epiph.umu.se)

Version: 7 Date: 18 April 2013

Author’s response to reviews: see over
Dear Editor

Find attached the manuscript titled “Easier said than done: applying the Ecohealth principles to a study of heavy metals exposure among indigenous communities of the Peruvian Amazon”.

This manuscript presents the application of the participatory approach Ecohealth to an epidemiological study of heavy metals exposure among indigenous communities in the Peruvian Amazon, in order to illustrate how knowledge generated from a participatory research process does not necessarily imply solving a public health problem.

We believe that the use of participatory approaches in epidemiological studies is gaining growing interest. Thus, the awareness and understanding of their attributes and challenges is necessary among the scientific community. In order to contribute to this purpose and assist other researchers working in similar contexts, we submit this original article and hope it will be considered for publication in your journal.

The four authors declare that they have no competing (financial) interests. They have read the manuscript and confirm that it is ready for submission. As suggested by the reviewers, the manuscript has been edited by a professional English-language editing service.

Yours sincerely

Cynthia Anticona, corresponding author
Cynthia Anticona MPH, PhD
Department of Public Health and Clinical Medicine, Epidemiology and Global Health
Umeå University S- 901 85 Umeå Sweden.

e-mail address: cynthiaanticona@hotmail.com, cynthiaanticona@yahoo.com